Author’s response to reviews

Title: Elevated serum levels of aminotransferases in relation to unhealthy foods intake: Tehran Lipid and Glucose Study

Authors:

Parvin Mirmiran (mirmiran@endocrine.ac.ir)

Zahra Gaeini (zahragn1992@gmail.com)

Zahra Bahadoran (zahrabahadoran@yahoo.com;z.bahadoran@endocrine.ac.ir)

Fereidoun Azizi (azizi@endocrine.ac.ir)

Version: 2 Date: 27 Jul 2019

Author’s response to reviews:

according to the comments;

1. Availability of data and materials section revised as following: The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

2. in Authors’ Contributions section, all authors referred by their initials names.

3. Clean manuscript provided and uploaded.